Author's response to reviews

Title: Assessment of the relationship between bacteriological quality of dug-wells, hygiene behaviour and well characteristics in two cholera endemic localities in Douala, Cameroon.

Authors:

Jane-Francis T.K. Akoachere (jakoachere@yahoo.com)
Lundi-Anne N.B. Omam (lundiannen@yahoo.com)
Thomas N. Massalla (thomasmasalla@yahoo.com)

Version: 2 Date: 28 March 2013

Author's response to reviews:

Ethical approval to collect data for this study was obtained from the Faculty of Health Sciences Institutional Review Board of the University of Buea (No. 2011-02-0159).